Author's response to reviews

Title: Efficacy and safety of a booster dose of influenza vaccination in solid organ transplant recipients: TRANSGRIPE 1-2, study protocol for a multicentre, randomized, controlled clinical trial.

Authors:

Juliana Martinez-Atienza (jumarati@gmail.com)
Clara M Rosso-Fernández (claram.rosso.sspa@juntadeandalucia.es)
Cristina Roca (cristinaroca85@gmail.com)
Teresa A Aydillo (teresa.aydillo@gmail.com)
Joan Gavalda (gavaldasantapau@gmail.com)
Asunción Moreno (amoreno@clinic.ub.es)
Jose M Montejro (josemiquelmontejro@hotmail.com)
Julian Torre-Cisneros (julian.torre.sspa@juntadeandalucia.es)
M Carmen Fariñas (mirfac@humv.es)
Jesus Fortun (fortunabete@gmail.com)
Nuria Sabé (nfsab@bellvitgehospital.cat)
Patricia Muñoz (pmunoz@micro_hggm.es)
Marino Blanes-Julia (blanes@ono.es)
Alejandro Suárez-Benjumea (suarezbenjumea@gmail.com)
Francisco López-Medrano (flmedrano@yahoo.es)
Pilar Perez-Romero (mperez-ibis@us.es)
Elisa Cordero (mariae.cordero.sspa@juntadeandalucia.es)
TRANSGRIPE 1-2 study group (transgripe@gmail.com)

Version: 7 Date: 24 July 2014

Author's response to reviews: see over
Trials Editorial Team  
University of Oxford, UK

Seville, 24th July 2014

Dear Mrs. Mc Gavin,

Thank you for your comments regarding our recently submitted manuscript, with reference number MS: 1715597979122432. On behalf of all the authors, please find enclosed the revised manuscript of the study protocol entitled: “Efficacy and safety of a booster dose of influenza vaccination in solid organ transplant recipients, TRANSGRIPE 1-2: study protocol for a multicentre, randomized, controlled clinical trial”. As suggested we have taken into account your comments and modified the manuscript accordingly. A point-by-point response follows:

1. Please ensure the title conforms to journal style for study protocol articles. The title should follow the format “__________: study protocol for a randomized controlled trial.”

   The title of the article has been slightly modified as indicated.

2. Please include all author affiliations and email addresses on your title page.

   All authors’ affiliations have been moved to the title page of the article. Only the corresponding author has provided her email address for correspondence.

3. Please move your funding information to the Acknowledgements section.

   Funding and sponsorship information has been moved to the acknowledgements section.

4. Please include a figure title and legend section after your reference list.

   Figure title and legend have been included.

We look forward to hearing from you.

Sincerely,

Clara M. Rosso Fernández, Ph.D. and M.D.
Virgen del Rocío University Hospital
Avda. Manuel Siurot s/n
41013 Seville (SPAIN)
Phone: +34 955 013414
Fax: +34 954 232992
Email: claram.rosso.sspa@juntadeandalucia.es